Author’s response to reviews

Title: Aerosol drug delivery to the lungs during nasal high flow therapy: an in vitro study

Authors:

Martin Wallin (mwal4000@uni.sydney.edu.au)
Patricia Tang (patricia.tang@sydney.edu.au)
Rachel Chang (yoon.chang@sydney.edu.au)
Mingshi Yang (mingshi.yang@sund.ku.dk)
Warren Finlay (finlay@ualberta.ca)
Hak-Kim Chan (kim.chan@sydney.edu.au)

Version: 1 Date: 03 Dec 2018

Author’s response to reviews:

We are grateful and we thank the reviewers for their constructive comments. We have detailed our responses and actions taken point-by-point in the document attached. Any additions or changes made in the manuscript are in red for easy viewing. The line numbers used in our response below refer to the clean version of the manuscript.